Author’s response to reviews

Title: Benefit in liver transplantation. A survey among medical staff, patients, medical students and non-medical university staff and students

Authors:

Christine Englschalk (christine.englschalk@campus.lmu.de;christine@englschalk.de)
Daniela Eser (Daniela.Eser@med.uni-muenchen.de)
Ralf Jox (ralf.jox@med.uni-muenchen.de)
Alexander Gerbes (Gerbes@med.uni-muenchen.de)
Lorenz Frey (Lorenz.Frey@med.uni-muenchen.de)
Derek Dubay (dubay@musc.edu)
Martin Angele (Martin.Angele@med.uni-muenchen.de)
Manfred Stangl (Manfred.Stangl@med.uni-muenchen.de)
Bruno Meiser (Bruno.Meiser@med.uni-muenchen.de)
Jens Werner (Chir_Direktion_AVT-Klinik@med.uni-muenchen.de)
Markus Guba (Markus.Guba@med.uni-muenchen.de)

Version: 3 Date: 28 Jan 2018

Author’s response to reviews:

METH-D-17-00071R2 Response letter

Dear Ms. Tian and Ms. Wright,

According to the editor comments, we have revised the manuscript.
1. In the section ‘Ethics approval and consent to participate’ we added a sentence that consent was assumed as a result of the completion of the survey and stated that this survey met local guidelines and included the name of the ethics committee.

3. In the section ‘Authors’ contributions’ we included that all authors read and approved the final version of the manuscript.

Thank you for receiving our revision.

Best regards,

Prof. Dr. Markus Guba, Christine Englschalk